%%%%%%%%%%%%%% Presentation Template %%%%%%%%%%%%%%%%%%%%
\documentclass[a4,12pt]{article}
\usepackage{epsfig}

\setlength{\topmargin}{3mm}
\setlength{\textheight}{229mm}
\setlength{\evensidemargin}{-3mm}
\setlength{\oddsidemargin}{-3mm}
\setlength{\textwidth}{154mm}

\thispagestyle{empty}

\begin{document}

%\input{helge_header}
\input{hopfentation_header}
{\scriptsize \bf 05/06}
\vspace{4ex}

\centerline{\Huge\bf Our solutions II}
\vspace{1cm}

\Large
\begin{itemize}

 \item Transformation to activation pattern; the signal in feature space
 \begin{itemize}
  \item Return signal from the tomograph algorithm
  	\begin{figure}[!ht]
	\begin{center}
	\includegraphics[width=14cm]{patfeat.eps}
	\end{center}
	\end{figure}  
  
  \item The transformation to a network activation pattern
  	\begin{figure}[!ht]
	\begin{center}
	\includegraphics[width=6cm]{pattern.eps}
	\end{center}
	\end{figure}  

 \end{itemize}

\vspace{0.5cm}

\item The Hopfield network 
  	\begin{figure}[!h]
	\begin{center}
	\includegraphics[width=10cm]{neuronex.eps}
	\end{center}
	\end{figure}  

\end{itemize}
\clearpage

\end{document}
